Accuracy of glomerular filtration rate (GFR) estimation using creatinine and cystatin C and albuminuria for monitoring disease progression in patients with stage 3 chronic kidney disease: prospective longitudinal study in a multi-ethnic population

eGFR-C Study



STUDY PROTOCOL: VERSION 4.0 2nd February 2017

Co-Sponsors:	East Kent Hospitals University NHS Foundation Trust and University of Birmingham
Chief Investigator:	Dr Edmund Lamb East Kent Hospitals University NHS Foundation Trust
Coordinating Centre:	Birmingham Clinical Trials Unit (BCTU)
Funder:	National Institute for Health Research Health Technology Assessment programme (HTA 11/103/01)
ISRCTN42955626	
IRAS Ref. No.:	135229
Main REC Ref. No.:	13/LO/1349

East Kent Hospitals University NHS





STUDY COMMITTEES AND CONTACT DETAILS

Study Management Group

Chief Investigator		
Dr Edmund Lamb	Consultant Clinical Scientist East Kent Hospitals University NHS Foundation Trust	
Co-Investigators		
Dr Jonathan Barratt	Consultant Nephrologist Leicester General Hospital Email: jonathan.barratt@uhl-tr.nhs.uk	
Professor Paul Cockwell	Consultant Nephrologist University Hospitals Birmingham NHS Foundation Trust Email: <u>paul.cockwell@uhb.nhs.uk</u>	
Professor R Neil Dalton	Professor of Paediatric Biochemistry Kings College London Email: <u>neil.dalton@kcl.ac.uk</u>	
Professor Jon Deeks	Professor of Biostatistics University of Birmingham and Birmingham CTU Email: j.deeks@bham.ac.uk	
Ms Tracy Pellatt-Higgins	Statistician University of Kent Email: <u>t.pellatt-higgins@kent.ac.uk</u>	
Professor Phil Kalra	Consultant Nephrologist Salford Royal NHS Foundation Trust Email: <u>philip.kalra@srft.nhs.uk</u>	
Professor Kamlesh Khunti	Professor of Primary Care Diabetes & Vascular Medicine University of Leicester Email: <u>kk22@leicester.ac.uk</u>	
Mrs Fiona Loud	Policy Director British Kidney Patient Association Email: <u>fiona.loud@live.co.uk</u>	
Dr Claire Sharpe	Clinical Senior Lecturer in Renal Medicine King's College London & King's College Hospital NHS Foundation Trust Email: <u>claire.sharpe@kcl.ac.uk</u>	
Dr Paul Stevens	Consultant Nephrologist East Kent Hospitals University NHS Foundation Trust Email: <u>pstevens@nhs.net</u>	
Dr Andrew Sutton	Associate Professor in Decision Modelling University of Leeds Email: <u>a.j.sutton@leeds.ac.uk</u>	
Professor Maarten Taal	Consultant Nephrologist Royal Derby Hospital Email: <u>maarten.taal1@nhs.net</u>	

Clinical Trials Unit (BCTU)	
Study Management Hugh Jarrett	Renal Team Leader Birmingham Clinical Trials Unit University of Birmingham Email: <u>h.jarrett@bham.ac.uk</u>
Ryan Ottridge	Study Co-ordinator Birmingham Clinical Trials Unit University of Birmingham Email: <u>r.ottridge@bham.ac.uk</u>
Adam Khan	Data Manager Birmingham Clinical Trials Unit University of Birmingham Email: <u>a.r.khan@bham.ac.uk</u>
Statistician Ms Alice Sitch	University of Birmingham Email: <u>a.j.sitch@bham.ac.uk</u>
Stud	y Steering Committee
Dr Charlie Tomson (Chair)	Consultant Nephrologist Freeman Hospital, Newcastle Email: <u>charles.tomson@nhs.net</u>
Dr Hugh Gallagher	Consultant Nephrologist St Helier Hospital, Carshalton, Surrey Email: <u>hugh.gallagher@esth.nhs.uk</u>
Professor Rafael Perera	Head of Medical Statistics Group Department of Primary Care Health Sciences University of Oxford Email: <u>rafael.perera@phc.ox.ac.uk</u>
Mr Nicholas Palmer	Head of Patient Support Services British Kidney Patient Association Email: <u>nicholas.palmer@britishkidney-pa.co.uk</u>
Dr Edmund Lamb (Chief Investigator)	Consultant Clinical Scientist East Kent Hospitals University NHS Foundation Trust Email: <u>elamb@nhs.net</u>

eGFR-C Study Office

For general protocol related queries and supply of study materials:

Birmingham Clinical Trials Unit (BCTU), Public Health Building, University of Birmingham, Birmingham B15 2TT

> Telephone: 0121 415 9134 Fax: 0121 415 9135

Email: eGFR-C@trials.bham.ac.uk

Website: www.birmingham.ac.uk/egfrc

Registration

Telephone: 0800 953 0274 (toll free in the UK, available 9am-5pm Monday – Friday)

Website: https://www.trials.bham.ac.uk/egfrc

Safety Reporting

Fax SAE Forms to: 0121 415 9135

This protocol describes the eGFR-C study only. The protocol should not be used as a guide for the treatment of patients not taking part in the eGFR-C study. The study will be conducted in accordance with the protocol, Good Clinical Practice (GCP). Every care has been taken in the drafting of this protocol, but future amendments may be necessary, which will receive the required approvals prior to implementation.

Chief Investigator and Co-Sponsor Signatures

The Chief Investigator and the Co-Sponsors have discussed this protocol. The Investigator agrees to perform the investigations and to abide by this protocol.

The Investigator agrees to conduct the study in compliance with the approved protocol, EU Good Clinical Practice (GCP), the UK Data Protection Act (1998), the Trust Information Governance Policy (or other local equivalent) and the Research Governance Framework (2005 2nd Edition; as amended).

eGFR-C Study Protocol Version 4.0, 2nd February 2017

Previous version: Version 2.0, 5th October 2016

The protocol has been approved by:

Chief investigator

Dr Edmund Lamb, Consultant Clinical Scientist East Kent Hospitals University NHS Foundation Trust

Signature

Date

Sponsor Representative

Dr Art Ationu East Kent Hospitals University NHS Foundation Trust

Signature

Date

For University of Birmingham sponsored trials, the sponsor will confirm approval of the protocol by signing the IRAS form and therefore a signature on the protocol is not required.

Principal Investigator Signature Page

Principal Investigator:

I have read and agree to the protocol, as detailed in this document. I agree to adhere to the protocol as outlined and agree that any suggested changes to the protocol must be approved by the Research Ethics Committee (REC).

I am aware of my responsibilities as an Investigator under the guidelines of Good Clinical Practice (GCP), the Declaration of Helsinki and the study protocol and I agree to conduct the study according to these guidelines and to appropriately direct and assist the staff under my control, who will be involved in the study.

eGFR-C Study Protocol Version 4.0, 2nd February 2017

Previous version: Version 2.0, 5th October 2016

Principal investigator

<insert name>

Signature

Date

Name of Institution

<insert name>

Each Principal Investigator should sign this page and return a copy to the eGFR-C Study Office

Table of Contents

Chie	f Investigator and Co-Sponsor Signatures4
Princ	cipal Investigator Signature Page5
List o	of Abbreviations
Sum	mary & Study Schema9
1	Introduction13
1.1	Background13
1.2	Clinical data17
1.3	Rationale for the study and assessment of risk18
1.4	Assessment and management of risk19
2	Study Design
3	Study Objectives
4	Selection of Participants21
4.1	Inclusion criteria21
4.2	Exclusion criteria21
5	Recruitment22
6	Study Procedures and Schedule of Assessments23
6.1	Screening procedures23
6.2	Informed consent procedure24
6.3	Withdrawal25
6.4	Registration procedures25
6.5	Baseline assessments26
6.6	Blinding of Test Results27
6.7	Assessment schedule
6.8	Study Duration29
6.9	Follow-up of participants beyond study closure
7	Study Procedures
7.1	Management of Participants
7.2	Laboratory procedures
8	Recording and Reporting of Adverse Events
9	Data Management and Quality Assurance

9.1		Confidentiality of Personal Data	
9.2	1	Long-Term Storage of Data	37
9.3		Data collection	
9.4		Data handling and analysis	
10	Arch	iving	40
11	Stati	stical Considerations	40
11.	1	Outcome Measures	40
11.	1.1	Main study	40
11.	1.2	Sub-study of patterns of disease progression	41
11.	1.3	Additional study of intra-individual biological variation	41
11.	2	Sample size	41
11.	3	Statistical analysis	42
11.	4	Interim analyses	47
11.	5	Final analyses	47
11.	6	End of study	48
12	Dire	ct Access to Source Data	
13	Ethic	cs Requirements	
14	Mon	itoring Requirement for the Study	
15	Fina	nce	
16	Inde	mnity	
17	Diss	emination and Publication	50
18	State	ement of Compliance	50
19	Refe	erences	51

List of Abbreviations

ACR	Albumin to creatinine ratio
AE	Adverse Event
AR	Adverse Reaction
CI	Chief Investigator
CKD	Chronic kidney disease
CKD-EPI	Chronic kidney disease-epidemiology consortium
CRF	Case Report Form
CVi	Within-individual biological variation
eGFR	Estimated glomerular filtration rate
GFR	Glomerular filtration rate
GCP	Good Clinical Practice
НТА	Health Technology Assessment
ISRCTN	International Standard Randomised Control Trial Number
MDRD	Modification of Diet in Renal Disease
NHS R&D	National Health Service Research & Development
NIHR	National Institute for Health Service Research
PI	Principal Investigator
PIS	Participant Information Sheet
MREC	Main Research Ethics Committee
RCV	Reference change value
SAE	Serious Adverse Event
SmPC	Summary of Product Characteristics
SSA	Site Specific Assessment
SMG	Study Management Group
SSC	Study Steering Committee

Summary & Study Schema

EXECUTIVE SUMMARY					
Title	Accuracy of glomerular filtration rate (GFR) estimation using creatinine and cystatin C and albuminuria for monitoring disease progression in patients with stage 3 chronic kidney disease (CKD)				
Acronym	eGFR-C				
Study design and methods	Multi-centre UK prospective longitudinal cohort study - 1300 participants will undergo baseline (month 0) and final (month 36) reference GFR, estimated GFR (eGFR) and urinary albumin-to-creatinine ratio (ACR) tests. Additionally they will provide ACR and eGFR tests at 6-monthly intervals. A subset of the cohort (n=375) will receive annual reference GFR tests.				
	Biological variability study – In a further sub-study 20 participants will undergo the reference test four times over four weeks.				
	Modelling of monitoring and cost effectiveness - Optimal monitoring regimens and cost-effectiveness will be assessed using data from the longitudinal study and the biological variability study.				
Total number of participants planned	1320				
Study duration per participant	36 months				
Accrual period	35 months				
Estimated total study duration	83 months				
Primary study objectives	 To estimate and compare the accuracy and precision of GFR- estimating equations based on the MDRD equation and three CKD-EPI equations using either creatinine or cystatin C or a combination of both in people with stage 3 CKD. To estimate the accuracy and precision of the GFR-estimating equations according to ethnic group (particularly Caucasians, African-Caribbean and South Asian), and baseline diabetes and proteinuria. To evaluate and compare how accurately these GFR-estimating equations reflect change in GFR over three years. To establish which GFR-estimating equation, together with ACR, or ACR alone, most accurately predicts those people that have progressive loss of kidney function (CKD progression). To estimate and model disease progression (decline in GFR or increase in ACR) according to ethnic group (particularly Caucasians, African-Caribbean and South Asian), and baseline diabetes and proteinuria 				

	 To compare the effectiveness and costs of monitoring strategies for identifying people that have progressive loss of kidney function (CKD progression) utilising different GFR-estimating equations and test schedules, accounting for differences in risk of progression.
Main inclusion criteria	 Aged 18 years or older Patients with stage 3 CKD (GFR 30-59 mL/min/1.73 m²), diagnosed using MDRD/CKD-EPI eGFR (at least two consecutive test results in this range at least 90 days apart, with the most recent test in the last 12 months) Written informed consent
Main exclusion criteria	 History of untoward reactions to iodinated contrast media or allergy to topical iodine Episode of acute kidney injury in previous 6 months (as defined by the Acute Kidney Injury Network criteria) Amputation of whole or part-limb Pregnant or breastfeeding Known current alcohol or drug abuse Kidney transplant recipient Any condition with an expected survival of less than study duration Inability to comply with study schedule and follow-up Inability to provide informed consent e.g. due to cognitive impairment

LAY SUMMARY

The best measure of kidney function is accepted to be the glomerular filtration rate (GFR). which measures the ability of the kidney to filter blood and is widely used in clinical practice. A low GFR suggests poor kidney function. An estimate of GFR can be obtained from a simple blood test. In 2006, the Department of Health recommended that all NHS laboratories in England should routinely estimate and report GFR using equations based on the measurement of a substance in blood called creatinine. Similar recommendations were made throughout the rest of the UK and internationally. Creatinine is normally filtered and excreted by the kidneys, but if the kidneys function is reduced its concentration in blood increases. Hence a high creatinine concentration in blood equates to poor kidney function (i.e. low GFR). Age, gender and race can also affect creatinine in the blood, so adjustments can be made to the way in which GFR is estimated. The introduction of GFR reporting as part of routine clinical practice in the UK, plus more widespread detection of kidney disease through testing for excess protein in urine (proteinuria /albuminuria), has increased the number of people diagnosed with chronic kidney disease (CKD): approximately one in seven of the UK population have been diagnosed with CKD using these methods. However, there are concerns regarding the accuracy of this testing and its ability to detect deterioration in kidney function. More accurate testing would enable better identification and monitoring,

ultimately improving outcomes for people with CKD, whilst reducing risks of misdiagnosis. Cystatin C, a small protein that can be measured in blood, has been proposed as an improved marker of GFR. However, cystatin C is approximately ten times more expensive to measure than creatinine.

This study will assess the accuracy of current and alternative tests of kidney function against a reference test (see below) in a large group of people (1300 from six centres) with moderate (stage 3) CKD, who will be followed over a three year period. It will include people with diabetes and proteinuria, and people of Caucasian, African-Caribbean and South Asian descent.

The study will recruit people with stage 3 CKD from hospital clinics and from GP surgeries at six major UK centres (Birmingham, Canterbury, Derby, London, Leicester, Salford). Participants will be given an initial (reference) GFR test when they enter the study with a second follow-up reference test three years later. The reference test involves injecting a small amount of iohexol, a non-radioactive iodine-containing substance widely used for this purpose, into a vein. Blood samples will be taken over the next 4 hours to see how quickly the iohexol disappears from the blood stream. The rate at which iohexol disappears is due to renal excretion (i.e. via urine) and is equivalent to the level of kidney function. Blood tests for monitoring kidney function, including testing for creatinine, cystatin C and urinary albumin, will be done every six months during the study period.

In the study, iohexol measured GFR will be accepted as the reference ('gold standard') measure of kidney function. Each GFR-estimating equation will be compared against it. The alternative estimated measures of GFR, derived from measuring substances (creatinine and cystatin C) in the blood, will be compared against the reference test. An important outcome is how much the reference test changes over the three years of the study, and how well the alternative measures reflect this change. We will also collect accurate test cost data for further analysis of cost-effectiveness (e.g. do the relative costs of the tests justify any change in practice due to improved performance of one test compared to another?).

Participants will need to provide informed consent; those who are unable to do so will not be enrolled in the study. The main risk is of allergy to iodine contained within the reference test. This is very rare and individuals with a known allergy to iodine will be excluded.

Study Schema Figure 1



1 Introduction

1.1 Background

Chronic kidney disease (CKD) is defined as abnormalities of kidney structure or function for \geq 3 months with implications for health. CKD is prevalent in the population in general:¹⁻³ the 2009 Health Survey for England identified approximately 13% of the UK population as having CKD.⁴ Most commonly it is identified using estimation of glomerular filtration rate (GFR) or detection of protein in the urine (albuminuria/proteinuria).

GFR is accepted as the best overall measure of kidney function and is central to diagnosis, staging and management of CKD. Ideally GFR would be measured using reference procedures which follow the clearance of an infused exogenous substance (e.g. inulin, ¹²⁵I-iothalamate, ⁵¹Cr-EDTA or iohexol⁵). However, these methods are cumbersome and impractical for general kidney disease detection and management. Estimation of GFR (estimated GFR [eGFR]) using equations based on serum creatinine with adjustments for age, gender and race are widely used as surrogate measures of GFR. Estimation of GFR on every blood creatinine request received by laboratories has been recommended by the Department of Health.⁶ In England, the National Institute for Health and Care Excellence (NICE) have made recommendations regarding which people should be tested for the presence of CKD (e.g. those with diabetes or hypertension) and have recommended that GFR should be estimated 6-monthly in people with stage 3 CKD (GFR 30-59 mL/min/1.73 m²),⁷ comprising approximately 6-7% of the overall UK population.^{3, 8} It is estimated that more than 50 million GFR estimates are produced by UK NHS laboratories every year.

The aim of disease detection is to identify and manage those most likely to progress to kidney failure and/or who are at high risk of morbidity and mortality. In addition to the accurate identification of CKD, the ability of tests to identify which individuals with CKD will have high risk (i.e. progressive or mortal) disease is a crucial issue. Whilst introduction of routine GFR estimations is generally deemed to have brought significant health advantages,⁹ many people with stage 3 CKD are not at increased risk of progressive disease and there are concerns that CKD detection using creatinine-based approaches may be identifying individuals who are at low risk and unlikely to benefit from active management (e.g. blood pressure lowering, use of inhibitors/angiotensin angiotensin converting enzyme receptor blockers, cardiovascular risk reduction).¹⁰ Recently, newer equations utilising cystatin C instead of, or in addition to, creatinine have been proposed. Endorsement of cystatin C testing in international guidance,¹¹ together with the increasing availability of cystatin C assays on large, automated laboratory test platforms will increase the pressure on NHS laboratories to introduce this test, which is significantly more expensive than creatinine testing. Given the high costs of cystatin C testing compared to creatinine, it is critical that its diagnostic accuracy and prognostic ability are carefully validated ahead of widespread introduction into the NHS.

The ability of tests to identify which individuals with CKD will have high risk (i.e. progressive or mortal) disease is seen as a crucial issue. A significant problem has been the ability of GFR-estimating equations to identify progression of kidney disease given the biological variability of its main determinant (serum creatinine). There have been no prospective studies of the ability of GFR estimating equations to monitor progression and no studies at all of GFR estimating equations incorporating cystatin C; there have also been no validations of GFR estimating equations these important issues.

Measuring GFR

Standard clearance of inulin, including urine collection, remains the 'gold-standard' method for GFR measurement, but few studies use this as the method is cumbersome and inulin is not easily measured. Most evaluations of GFR equations have used radiolabelled plasma clearance methods which are assumed to be closely related to inulin clearance. Radiolabelled iothalamate plasma clearance was the method used for developing the Modification of Diet in Renal Disease (MDRD) Study ¹² and Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI)¹³ GFRestimating equations (see below), whilst the CKD-EPI equation validation dataset used a variety of reference GFR methods including iohexol.¹³ Although regarded as the reference approach for the assessment of kidney function, it is increasingly appreciated that non-inulin plasma clearance methods are not all equivalent.¹⁴ Furthermore, as with any physiological measurement, GFR has an intrinsic biological variability, understanding of which is critical to the appreciation of diseaserelated change. Using a variety of reference markers, values (coefficient of variation, CV%) ranging between 5.5% and 11.6% have been reported for the biological variation of GFR.¹⁵⁻²¹ However, most of these estimates were not derived using classical biological variation studies as described by Fraser and Harris.²²

Estimating GFR

The MDRD Study equation (MDRD_{creatinine}), which estimates GFR adjusted for body surface area (BSA), was originally developed in 1999.¹² A simplified ('4-variable') version of the equation which requires knowledge only of serum creatinine concentration, age, gender and race (black or white) was later published²³ and

subsequently re-expressed for use with a standardised serum creatinine assay.²⁴ Generally, the MDRD equation has been seen to perform better, and offer practical advantages, over other GFR equations that had been used previously. Its use has been endorsed by national professional healthcare organisations²⁵ including those in the UK.⁷ However, accuracy of the equation is sub-optimal. In the CKD field, accuracy of GFR estimating equations is commonly expressed as the P₃₀, the percentage of eGFR values within 30% of 'true' GFR. This metric captures aspects of both imprecision (measurement error) and bias (systematic over- and/or underestimation). Reported P₃₀ values for the MDRD equation typically range between 73% to 93%.²⁶

The MDRD equation has also been criticised on the basis that it significantly underestimates GFR (particularly in individuals with GFR greater than 60 mL/min/1.73 m²) and has poor precision.¹³ A revised equation, the Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI_{creatinine}) equation has been published and is thought to partially address this issue, producing less biased estimates of GFR at higher levels of kidney function,¹³ although reportedly less accurate estimates as GFR falls below 60 mL/min/1.73 m².²⁶ P₃₀ values for the CKD-EPI_{creatinine} equation are slightly superior to those of the MDRD equation in studies that have undertaken a head-to-head comparison.²⁶

Cystatin C has been proposed as an improved marker of GFR compared to creatinine.^{27, 28} Recently, the CKD-EPI Collaboration have published two further CKD-EPI equations; one based on cystatin C (CKD-EPI_{cystatin C}) and one using both cystatin C and creatinine (CKD-EPI_{creatinine and cystatin C}).²⁹ Members of the current study group have recently published the only independent validation to date of these latter equations.³⁰

Estimating GFR in British ethnic minority populations

People from South Asian and African-Caribbean backgrounds are at 3 to 5-fold increased risk of developing established renal failure compared to Caucasians.^{31, 32} They are also at greater risk of complications from diabetes and high blood pressure than the rest of the population. Whilst GFR-estimating equations have been validated in African-Caribbean communities from North America³³ and endemic Asian populations,³⁴⁻³⁹ there is no independent validation in British African-Caribbean populations and no data at all amongst Asians from the Indian sub-continent. Although it is often assumed that progression is higher in black and Asian⁴⁰ ethnic groups compared to whites, this remains unproven.^{41, 42}

Progression of kidney disease

There is no consistent definition of what constitutes renal progression in the literature. Many studies have used a doubling of serum creatinine, corresponding to an approximate halving of GFR, as an end-point defining progression, but this is insufficiently sensitive to be useful in clinical practice. Kidney Disease Improving Global Outcomes (KDIGO) have defined progression as a move to a higher disease category (e.g. stage 3A [GFR 45-59 mL/min/1.73 m²] to stage 3B [GFR30-44 mL/min/1.73 m²]) confirmed by a fall in GFR of greater than or equal to 25% (e.g. a decline from 60 mL/min/1.73 m² to less than 45 mL/min/1.73 m²), an increase in albuminuria, or a greater than 10%/year decline in GFR (e.g. a decline from 60 mL/min/1.73 m² to less than 54 mL/min/1.73 m² in one year).¹¹ NICE defined progression as a decline in GFR of more than 5 mL/min/1.73 m²/year, or more than 10 mL/min/1.73 m²/5 years.⁷

Progression is not necessarily common even amongst people with known CKD e.g. amongst people with stage 3 CKD only 1.3% progressed to stage 5 CKD (established renal failure, typically requiring dialysis or transplantation) over 5 years.⁴³ Amongst community dwelling older (greater than 65 years) adults with stage 3 CKD, Hemmelgarm et al. reported mean decline of GFR of 3.6 mL/min/1.73 m²/year and 2.8 mL/min/1.73 m²/year respectively in male and female subjects with diabetes and somewhat lower values amongst subjects without diabetes (1.9 mL/min/1.73 m²/year and 1.1 mL/min/1.73 m²/year amongst males and females respectively).⁴⁴ In the REIN study, proteinuric (greater than 1 g/24 h) non-diabetic subjects with GFRs in the approximate range 30-50 mL/min/1.73 m² showed a decline of GFR of 7.0 mL/min/1.73 m²/year with slightly lower values being observed in those receiving renin-angiotensin-aldosterone system (RAAS) blockade.⁴⁵

In East Kent, of 4506 people in primary care with baseline eGFR 30-59 mL/min/1.73 m^2 , 18% had a fall in eGFR of over 10% in a 1 year period (SEiK study, Paul Stevens unpublished data). However, apparent increases in GFR were equally common. What is not known is whether such increases are real, or whether they reflect imprecision and inaccuracies of the creatinine-based estimate of GFR that was used.

There are some data, mainly restricted to small studies in people with diabetes, describing disease progression in terms of decline in reference GFR measurements.^{46, 47} Generally, disease progression in people with diabetes has been described as following a broadly linear decline, being influenced by blood pressure and albuminuria and ameliorated by antihypertensive medication/RAAS

blockade.^{46, 48, 49} A similar pattern has been observed using estimated rather than measured GFR.⁵⁰

1.2 Clinical data

Identifying progressive kidney disease

A significant problem is the ability of GFR-estimating equations to identify progression of kidney disease against background change in GFR (i.e. that due to 'normal' ageing; commonly cited as approximately 1 mL/min/1.73 m²/year) given the biological and measurement variability of both reference and estimated GFR. The intra-individual variation (CV) of the main determinant (serum creatinine) of eGFR has been reported as 4.3%⁵¹ to which should be added intralaboratory imprecision (CV_A) of approximately 3.0%.²⁵ On this basis, the critical difference or reference change value (RCV) for serum creatinine is 13% (i.e. this is the difference that can be considered 'real' with 95% probability). The power function in the MDRD equation (-1.154, Table 1) increases the impact of CV_1 to an average of 5.4%. Consequently the RCV for eGFR derived using the MDRD equation becomes 14.4%. As an example of this, in an individual a GFR of 60 mL/min/1.73 m² will need to fall below 51 mL/min/1.73 m² before it can be considered a significant decrease.* Some,^{52, 53} although not the majority⁵⁴⁻⁵⁷ of data suggests that the biological variation of serum cystatin C is greater than that of creatinine. If this were the case then it would clearly impact on the ability of cystatin C based GFR-estimating equations to detect changes in true GFR vis-à-vis serum creatinine.

GFR changes of the order discussed above exceed the limit that most nephrologists would consider a clinically insignificant change. However, there are no prospective longitudinal data assessing the relative abilities of GFR-estimating equations to detect change in underlying 'true' GFR. One recent study has addressed the accuracy of GFR estimating equations compared to ¹²⁵I-iothalamate measured GFR over time in people with kidney disease.⁵⁸ The authors concluded that GFR estimating equations accurately reflected changes in measured GFR over time. The study was robust (3,532 participants with CKD followed for a mean of 2.6 years), but was retrospective, and did not include data derived using cystatin C.⁵⁸ Observational data suggests that for identification of progressive CKD, the combination of eGFR using cystatin C alone, then the combination of ACR and eGFR using creatinine, and finally eGFR using creatinine alone.⁵⁹⁻⁶¹ The combined use of

^{*} The RCV % is constant, but clearly gives rise to differences in terms of mL/min/1.73 m² depending on the starting point. e.g. a 14.4% fall at 30 mL/min/1.73 m² only requires a decrease of 4.3 mL/min/1.73 m², whereas a 14.4% fall at 60 mL/min/1.73 m² requires a decrease of 8.6 mL/min/1.73 m².

cystatin C and creatinine in a GFR estimating equation, which is claimed to be less influenced by ethnicity, as a predictor of progression has not been tested.

1.3 Rationale for the study and assessment of risk

Whilst there is significant literature describing the accuracy of creatinine-based eGFR against reference methods (see Section 2.1 above), there are no studies addressing the ability of GFR estimating equations, including those incorporating cystatin C, to detect change in GFR. Furthermore, there are no data addressing the accuracy of these equations in British ethnic minority populations. This study will assess whether eGFR using either creatinine or cystatin C, or a combination of both, is superior at detecting changes in GFR as measured by a reference GFR method. The utility of baseline eGFR and urinary ACR to predict which people are likely to show progressive kidney disease will also be tested. We have chosen plasma iohexol clearance as the reference measure of GFR for this study because it is equivalent to inulin clearance, is widely used in clinical and research practice, is non-radioisotopic, can be measured accurately and precisely, and is cheap.^{16, 62, 63} We have found that the test can be accurately and safely delivered by research nurses and encountered no adverse events amongst 400 recently studied older people.³⁰ We have chosen to study the CKD-EPI and MDRD equations because these are the only GFR estimating equations anchored to both creatinine and cystatin C reference methodology, and therefore likely to generate data that will be valid in perpetuity.

The study population will be a large cohort of people with stage 3 CKD including subjects of African-Caribbean and South Asian ethnicity and including subjects with diabetes and proteinuria. This will build on the findings of previous research, but using a prospective design with regular reference GFR measurements: the impact of medication on disease progression will be estimated and included in the model. The model will be used to define optimal sampling times for high and low risk participants, and other significant subgroups defined by the model. We will also use a classical study design to establish the intra-individual biological variability of both reference and estimated GFR: this information will be used as one of the handles in defining progression and assessing the ability of GFR-estimating equations to detect it. A simple economic evaluation of the relative costs of the different tests will be included to enable cost-effectiveness modelling.

1.4 Assessment and management of risk

The assessment and management of risk is detailed in the separate eGFR-C Risk Assessment document. An on-going evaluation of risk will continue throughout the study.

The main risk to participants in this study is from the use of iohexol (Omnipaque 240). Iohexol is a non-ionic contrast agent widely used in radio-imaging procedures. Very rarely anaphylactoid reactions to iohexol are reported. Reported incidences of death from the use of water-soluble intravascularly administered contrast agents, including johexol, range from 6.6 per 1 million (0.00066 percent) to 1 in 10,000 (0.01 percent). Deaths are predominantly due to cardiac arrest. Isolated reports of hypotensive collapse and shock are also found in the literature. The incidence of shock is estimated to be 1 out of 20,000 (0.005 percent) patients. Reports of common side effects (e.g. sensation of warmth and pain upon injection) appear to be less common with iohexol than with other contrast agents (Manufacturer's product information, GE Healthcare Inc). Other less severe adverse reactions include nausea, urticaria and brochospasm. The risk of an adverse reaction to iohexol may be mitigated by the use of lower dose. For example, the dosage recommended for use in adults for contrast enhanced computed tomography studies is between 120 mL and 250 mL. In the present study, participants will receive 5 mL of Omnipaque 240. The procedure will be undertaken in a hospital setting with full resuscitation facilities available.

2 Study Design

The study will evaluate the performance of GFR-estimating equations, including novel equations incorporating cystatin C, in assessing and monitoring measured GFR in people with stage 3 CKD. The data will be analysed to assess the impact of ethnicity, proteinuria and diabetes on equation performance. There are three studies that will run concurrently:

1. **Main study:** This will comprise a 3-year longitudinal prospective cohort study using 6-monthly estimates of GFR, and baseline and final reference GFR values to assess and compare the accuracy and precision of each estimate of GFR and change in GFR. The study will be undertaken at 6 centres (Birmingham, Canterbury, Derby, Leicester, London and Salford) and will recruit 1300 subjects. Investigators will be blinded to the results of the reference tests. We will include assessments in high risk subgroups and ethnic groups.

2. **Sub-study of patterns of disease progression**: 375 participants (with an ideal split of 125 Caucasian, 125 African-Caribbean and 125 South Asian subjects to use

as a guideline to recruitment although overall sub-study numbers will be considered more important than any particular ethnic group) stratified within ethnicity by risk of disease progression (approximately 50% with diabetes and/or proteinuria and 50% with neither diabetes nor proteinuria) will undergo additional testing to that described in (1), with a reference GFR measurement each year over the 3 year study period. The study will enable a model of disease progression to be developed based on reference GFR measurement enabling optimal monitoring frequencies in the highrisk cohort to be defined. This number of subjects should provide a range of values over the main factors considered to influence disease progression and allow assessment of covariates in the statistical model. Further assessment of covariates will be performed by combining the data from this sub-study with the main study.

3. **Study of intra-individual biological variability:** At the Canterbury centre, a study will be undertaken to define the normal biological variability of the reference GFR test, in addition to the eGFR and urinary ACR tests. An additional twenty people with stage 3 CKD will undergo four iohexol reference measures of GFR in four successive weeks, with standardisation for time of day (morning after a light breakfast) and day of week. It is not anticipated that these subjects will also participate in the main study, although there would be no objection to them so doing.

3 Study Objectives

The aims of the study are:

- 1. To estimate and compare the accuracy and precision of GFR-estimating equations based on the MDRD equation and three CKD-EPI equations using either creatinine or cystatin C or a combination of both in people with stage 3 CKD.
- 2. To estimate the accuracy and precision of the GFR-estimating equations according to ethnic group (particularly Caucasians, African-Caribbean and South Asian), and baseline diabetes and proteinuria.
- 3. To evaluate and compare how accurately these GFR-estimating equations reflect change in GFR over three years.
- 4. To establish which GFR-estimating equation, together with ACR, or ACR alone, most accurately predicts those people that have progressive loss of kidney function (CKD progression).
- 5. To estimate and model disease progression (decline in GFR or increase in ACR) according to ethnic group (particularly Caucasians, African-Caribbean and South Asian), and baseline diabetes and proteinuria.
- 6. To compare the effectiveness and costs of monitoring strategies for identifying people that have progressive loss of kidney function (CKD progression) utilising

different GFR-estimating equations and test schedules, accounting for differences in risk of progression.

4 Selection of Participants

Participants who potentially fulfil the inclusion criteria for this study must have their eligibility confirmed by medically qualified personnel with access to and a full understanding of the potential participant's medical history. If eligibility has been assessed and documented by medically qualified personnel, then the process of receiving informed consent may be delegated as appropriate and as documented on the eGFR-C Delegation and Signature Log.

4.1 Inclusion criteria

- Aged 18 years or over
- Patients with stage 3 CKD (GFR 30-59 mL/min/1.73 m²) as defined internationally,¹¹ diagnosed using MDRD/CKD-EPI eGFR (at least two consecutive test results in this range at least 90 days apart, with the most recent test in the last 12 months)
- Written informed consent

4.2 Exclusion criteria

- History of untoward reactions to iodinated contrast media or allergy to topical iodine
- Episode of acute kidney injury in previous 6 months (as defined by the Acute kidney Injury Network criteria)
- Amputation of whole or part-limb
- Pregnant or breastfeeding
- Known current alcohol or drug abuse
- Kidney transplant recipient
- Any clinical condition with an expected survival of less than study duration
- Inability to comply with study schedule and follow-up
- Inability to provide informed consent e.g. due to cognitive impairment

5 Recruitment

A flowchart of the recruitment process is shown in the Study Schema (**Figure 1**) together with the investigation schedule.

Participants will be recruited to the main study from six centres and their dependent Participant Identification Centres (approximately n=220 each) as follows:

- 1. Birmingham: primary and secondary care
- 2. Canterbury: primary and secondary care
- 3. Derby: primary and secondary care
- 4. Leicester: primary and secondary care
- 5. Salford: primary and secondary care
- 6. London: Kings College Hospital: primary and secondary care

Recruitment in secondary/tertiary care will be from CKD clinics. Potential participants will be identified by the research team at each of the recruiting centres (e.g from individual renal unit databases or other local registries). The recruitment procedure is described in more detail in section 7.2.

In the primary care centres, letters will be sent to GPs inviting them to participate in the study. GPs who agree to participate in the study will then invite potential participants by mail. People who wish to participate in the study will contact the research nurse who will then recruit them into the study as described in section 7.2. If necessary, the research nurse will contact the GP to request verification of the eligibility information on the basis of which participants were invited. The research nurse will only request that the GP provides a "yes" or "no" to the eligibility verification, and not request additional information.

There will be no selection of participants to the main study based on ethnicity.

It is expected that most participants in the sub-study of patterns of disease progression will be recruited from three centres (n = approximately 125 each site). All centres will be free to recruit participants from any of the targeted ethnic groups into the sub-study, but based on the local populations sub-study recruitment is expected to follow the below pattern:

- 1. Birmingham: approximately 75% South Asian and African-Caribbean and 25% Caucasian participants
- 2. Leicester: approximately 50% South Asian and 50% Caucasian participants
- 3. London: Kings College Hospital: approximately 75% African-Caribbean and 25% Caucasian participants

Individuals will be stratified within ethnicity by risk of disease progression (at each of the three sites approximately 60 with diabetes and/or proteinuria and 60 with neither diabetes nor proteinuria). Participants will be identified from primary or secondary care as above. All consecutive patients will be invited to participate in the sub-study until the above ethnic stratification is achieved.

Participants (n=20) will be recruited to the study of intra-individual biological variability at Canterbury. Suitable consecutive participants will be invited.

6 Study Procedures and Schedule of Assessments

6.1 Screening procedures

Eligibility will be assessed against the inclusion and exclusion criteria and participants will then be identified as described above.

In the primary care centres, GPs who agree to participate in the study will invite potential participants by mail. A research assistant will visit consenting practices to help staff identify eligible people from the CKD register, for example using READ code searches. MIQUEST (Morbidity Information QUery and Export Syntax) software will be used to extract an up-to-date dataset. The research assistant will produce stamped addressed invitation letters to eligible people for signature by the GP. The invitation will be sent out along with the patient information sheet. Potential recruits will be given a dedicated study phone line to use to indicate willingness to participate. People who wish to participate in the study will contact the research nurse who will then discuss the study with them as described above: assuming they remain willing to participate they will be given an appointment to attend the hospital when signed informed consent will be obtained (see below).

Adequate time will be given for consideration by the participant before taking part. In primary care it will be left to the potential participants to contact the research nurse at their convenience (see above). Assuming they are willing to participate, they will be given an appointment to attend the hospital (and instructions relating to test preparation) for consent, registration and the baseline assessments to be undertaken (see section 7.4). Potential participants in secondary/tertiary care, identified by the research team at each of the recruiting centres (e.g. from individual renal unit databases or other local registries) will be invited by mail. In some cases the research nurse or participant's clinician may introduce the study to the participant before providing them with the invitation letter and patient information sheet. The invitation will be sent out or provided along with the patient information before being contacted by the research nurse to discuss the study with them and to see if they are willing to participate in the study. Assuming they remain willing to participate

they will be given an appointment to attend the hospital when signed informed consent will be obtained (see below).

6.2 Informed consent procedure

Eligibility should be assessed and documented by a clinician and then the process of obtaining written informed consent may be delegated as appropriate (to a suitably trained member of the local research team). This must be clearly documented on the eGFR-C Delegation and Signature Log.

Potential participants will initially be provided with a written Participant Information Sheet (PIS) (i.e. the current Main Research Ethics Committee (MREC) approved version which should be on appropriately headed paper) and a covering letter explaining the study to them and inviting them to participate in the study. They will have time to consider the study and decide whether or not they wish to take part, and to discuss the study with their family and friends if they would like to. If the potential participant has any questions or queries about the study during this time they will have the opportunity to discuss the study with the research nurse. The potential participant will then either be contacted by, or will contact, a member of research team at their local centre. If they remain willing to participate in the study they will be offered an appointment to attend clinic. At this clinic appointment potential participants will have plenty of time to discuss the study further and to have any questions that they may have about the study answered. The complex nature of the study and the requirement to spend approximately five hours at the hospital for the reference test, on between two and four occasions, will be carefully explained. The investigator or designee will explain that there is no obligation for a patient to enter the study, that study entry is entirely voluntary, and that it is up to the patient to decide whether or not they would like to join. It will also be explained that the patient can withdraw at any time during the study, without having to give a reason and that their decision will not affect the standard of care they receive. Translated material and translators will be used for non-English speakers. Where possible there will be an independent translator made available to ensure that participants are fully informed and supported to make decisions about enrolment in the study.

At the secondary care centres reasons for non-participation will be recorded if the information is volunteered.

At the first appointment (time zero) the research nurse will go through the registration form including the eligibility checklist. Assuming the patient is eligible they will be asked to sign a separate consent form and will be registered into the study (see section 7.3). Informed consent will be obtained before any study-related procedures are undertaken. A copy of the signed informed consent form will be given to the participant. The original signed form will be retained at the study site in the

Investigator Site File and a copy placed in the medical notes. A copy will also be sent to the eGFR-C Study Office.

If new safety information results in significant changes in the risk/benefit assessment, the consent form and PIS will be reviewed and updated as necessary. Participants will be re-consented if appropriate.

When potentially interested participants of the ethnicity described in Section 5 attend clinic for baseline assessment, they will be asked if they would be willing to be involved in the 375 patient sub-study of patterns of disease progression which will involve additional reference (iohexol) GFR at 12 and 24 months as detailed in the Participant Information Sheet. Consent to join the 375 patient sub-study will be documented on the Consent Form.

When potentially interested participants attend clinic for baseline assessment at Canterbury they will be asked if they would be willing to be involved in the 20 patient study of intra-individual biological variability involving four reference (iohexol) GFR and blood tests over four weeks. Interested participants will be given a separate information sheet to consider detailing involvement in the study and, if willing to participate, will sign a separate Consent Form to document their consent to involvement in the 20 patient biological variation study.

6.3 Withdrawal

Participants may withdraw at any time during the study if they choose not to continue or if their clinical team feel that continued participation in the study is inappropriate.

Full details of the reason(s) for withdrawal should be recorded on the Case Report Forms (CRFs) if healthcare professional-initiated, otherwise a simple statement reflecting participant preference will suffice. Participants who withdraw from study testing, but continue with on-going follow-up and data collection should be followedup in accordance with the protocol, unless the participant elects not to be included.

6.4 Registration procedures

After all eligibility criteria have been confirmed and informed consent has been received, the participants can be registered into the study.

Telephone and online registration

Participants can be entered into the study via a secure 24 hour internet based registration service (<u>https://www.trials.bham.ac.uk/egfrc</u>). Each researcher will be provided with a unique log-in username and password in order to access the online

system. Online registration is available 24 hours a day, 7 days a week, apart from short periods of scheduled maintenance and occasional network problems.

Registration Forms will be provided to investigators and should be completed and used to collate the necessary information prior to registration. All questions and data items on the Registration Form must be answered before a Study Number can be given. If data items are missing registration will be suspended but can be resumed once the information is available. Only when all eligibility criteria and baseline data items have been provided will a Study Number be allocated. A confirmatory email will be sent to the local Principal Investigator and the named research nurse.

With the participant's prior consent their General Practitioner (GP) should be informed of their study participation.

Back-up registration

If the internet based registration service is unavailable for an extended period of time, a back-up paper registration will also be available at the BCTU. In this instance, investigators should ring the BCTU registration service, available 9-5 Monday - Friday (telephone number **0800 953 0274**).

6.5 Baseline assessments

Participants will attend hospital in the morning having been advised to avoid eating a meal with high meat or fish content after 10 pm the evening before.^a After obtaining informed consent and registering the patient in the study the clinical and investigational baseline assessments will be undertaken.

- A clinical (including cardiovascular) and drug history will be recorded using a standardised questionnaire: information on ethnicity will be gathered using a modified version of the 2011 UK Census Questionnaire
- Height will be measured to the nearest 0.1 cm with a rigid stadiometer
- Body weight will be measured in light indoor clothing to the nearest 0.1 kg
- Waist circumference will be recorded to the nearest 0.1 cm at the mid-point between the lower costal margin and the level of the anterior superior iliac crest

^a N.B. participants will be asked about meat/fish consumption, but if affirmative this should not be a barrier to proceeding with the baseline assessments. The request to avoid meat/fish consumption after 10 pm the evening before the study visit applies to all study visits.

- Hip circumference will be recorded to the nearest 0.1cm, from the widest point of the hips and the maximum protrusion of the gluteal muscles
- Brachial blood pressure will be measured as recommended by the British Hypertension Society (<u>http://www.bhsoc.org/how_to_measure_blood_pressure.stm</u>) three times in the sitting position using standardised Omron M7 digital sphygmomanometers (Omron Healthcare, Milton Keynes, UK). The second and third blood pressure readings will be recorded and the average calculated by the study database
- Blood (12 mL) will be taken for serum creatinine and cystatin C measurement and sample storage (see below)
- A urine sample will be collected for ACR
- Further aliquots of serum/plasma and urine will be stored at -80°c for potential analysis of future markers. These may include, but not be limited to, albumin, asymmetric dimethylarginine, beta trace protein, bone alkaline phosphatase, B-type natriuretic peptide, calcium, cholesterol, clusterin, Creactive protein, growth differentiation factor 15, hepatocyte growth factor, 1,25-dihydroxyvitamin D, fibroblast growth factor 23, fibulin-1, 25hydroxyvitamin D, interleukin-18, insulin-like growth factor-binding protein 7, kidney injury molecule-1, matrix gla protein, neutrophil gelatinase associated parathyroid hormone, phosphate, tissue inhibitor lipocalin, of metalloproteinases-2, symmetric dimethylarginine, total alkaline phosphatase, trefoil factor-3, troponin I and T. Refer to the laboratory manual for details of aliquot preparation and storage.
- An iohexol reference GFR measurement will be undertaken in addition to estimation of GFR using four GFR estimating equations

6.6 Blinding of Test Results

Results of tests undertaken specifically for the purposes of the study (including the iohexol reference GFR measurement, and the central laboratory serum creatinine and cystatin-C eGFR measurements) will not be made available to treating clinicians and participants whilst the study is ongoing and therefore will not influence patient management. There is no possibility of releasing results to clinicians because they are being tested in batches throughout the study. The exception to this will be a reference or estimated GFR result of less than 15 mL/min/1.73 m², which will be notified to the Chief Investigator. Study samples will be labelled with study ID, and tested blinded to clinical information and results of all previous test results. Clinicians will have access to standard laboratory tests as per usual practice.

6.7 Assessment schedule

	Screening	V1	V2	V3	V3	V5	V6	V7
		M1	M6	M12	M18	M24	M30	M36
		(consent, registration and baseline assessment)						
Identify suitable participants, review inc. / exc. criteria. Explain study procedures. Participant information sheet and cover letter to be sent potential participants.	x							
Contactsecondarycareparticipantsafter1ordaystodeterminewillingnesstoparticipate.Primary careparticipantswillcontactstudycentreif willingtoparticipate.Offerfollow-upappointment.secondarysecondarysecondary	x							
Confirm eligibility, consent, register in study		х						
Demographic and medical history, anthropometric and clinical assessment		x		(x)		(x)		x
Baseline and final blood tests (serum creatinine, cystatin C) plus aliquot storage)		x						x
Monitoring blood tests (serum creatinine, cystatin C) plus aliquot storage)			x	x	x	x	x	
Urine test (ACR)		х	x	х	х	х	x	х
Iohexol (GFR) clearance test		х		(x)		(x)		х

Table 1: Schedule of visits (V) and assessments by month (M)

Note 1: schedule for sub-studies of disease progression and biological variation will differ slightly from above. In the sub-study of disease progression participants will have additional iohexol (GFR) clearance tests at 12 and 24 months (x). In the study of intra-individual biological variation participants will have additional iohexol (GFR) clearance, blood (serum creatinine, cystatin C) and urine (ACR) tests at 2, 3 and 4 weeks and will then exit the study.

Note 2: visits should take place and samples be collected within \pm one month of the planned date according to the study schedule (e.g. the V2 visit and samples should occur between 5 and 7 months after study commencement). If the visit will occur outside of these limits, the study office should be contacted.

Note 3: all iohexol assessment visits should start before midday with meat/fish having been avoided since 10pm the evening before. 6 monthly follow-up visits can take place at any point in the day although the participant should again avoid meat/fish since 10pm the evening before.

Note 4: to avoid meat/fish and similar high protein intake, suggested permitted foods will be listed in the study manual.

6.8 Study Duration

The recruitment period will last 35 months. The follow-up phase of the study will cease when the last participant has completed the 36 month final iohexol GFR clearance test.

6.9 Follow-up of participants beyond study closure

The study will include optional consent to allow future linkage to patient data available in NHS routine clinical datasets, including primary care data (e.g. CPRD, THIN, QResearch), secondary care data (Hospital Episode Statistics; HES) and mortality data from the Office of National Statistics (ONS) through The Health and Social Care Information Centre and other central UK NHS bodies. The consent will also allow access to other new central UK NHS databases that will appear in the future. This will allow us to extend the follow-up of patients in the study and collect long-term outcome and health resource usage data without needing further contact with the study participants. This will facilitate future studies linking markers of kidney function testing measured during the study period to longer-term outcomes in participants.

7 Study Procedures

7.1 Management of Participants

Throughout the study, the participants will be managed according to their usual care as determined by their local clinical staff. None of the study test results will be available to participants or treating clinicians whilst the study is ongoing, and thus they will not alter any management decisions.

Reference standard test

GFR will be measured in participants using an iohexol clearance method. A 5 mL bolus of Omnipaque 240 (518 g/L iohexol corresponding to 240 g/L of iodine, GE Healthcare www.gelifesciences.com) followed by 10 mL of normal saline will be injected into the antecubital vein. Blood samples will be collected at 5, 120, 180 and 240 minutes after injection. Exact time of the samples in relation to the bolus injection will be accurately recorded. Participants will be allowed free access to fluids during the collection procedure, but will be asked to refrain from excessive exercise and protein intake (i.e. meat and fish). Selected food as detailed in the study manual will be permitted.

Study tests

Blood samples will be collected using standard venepuncture and phlebotomy procedures including the use of a tourniquet. Blood will be collected in appropriate gel separating tubes following the manufacturer's recommended order of draw. The random urine sample will be taken into a plain sterilin pot.

Sample storage and transport

Samples will be transported to the local laboratory, where plasma/serum will be separated by centrifugation. Aliquots of serum/plasma and urine will then be stored at -80°C within 6 hours of venepuncture pending transportation to the central laboratories (St. Thomas's [iohexol, ID-MS creatinine] or Canterbury [enzymatic creatinine, cystatin C, ACR] depending on analyte) for analysis. See study manual for further details.

7.2 Laboratory procedures

A study manual documenting all laboratory procedures will be created. All analyses will be undertaken in accredited laboratories by staff registered with the Health Care Professions Council. GFR will be estimated using the simplified ID-MS traceable version of the MDRD equation and the three CKD-EPI equations (Table 2).

Abbreviation	GFR equation expressed as a single equation
MDRD _{creatinine} ²⁴	175 x Scr ^{-1.154} x age ^{-0.203} x 0.742 [if female] x 1.212 [if black]
CKD- EPI _{creatinine} ¹³	141 x min(Scr/ κ , 1) ^{α} x max(Scr/ κ , 1) ^{-1.209} x 0.993 ^{Age} x 1.018 [if female] x 1.159 [if black], where Scr is serum creatinine, κ is 0.7 for females and 0.9 for males, α is -0.329 for females and -0.411 for males, min indicates the minimum of Scr/ κ or 1, and max indicates the maximum of Scr/ κ or 1
CKD-EPI _{cystatin} c ²⁹	133 x min(Scys/0.8, 1) ^{-0.499} x max(Scys/0.8, 1) ^{-1.328} x 0.996^{Age} x 0.932 [if female] where Scys is serum cystatin C, min indicates the minimum of Scr/ κ or 1, and max indicates the maximum of Scr/ κ or 1.
CKD-EPI _{cystatin-} 29 creatinine	135 x min(Scr/κ, 1) ^α x max(Scr/κ, 1) ^{-0.601} x min(Scys/0.8, 1) ^{-0.375} x max(Scys/0.8, 1) ^{-0.711} x 0.995 ^{Age} x 0.969 [if female] x 1.08 [if black] where Scr is serum creatinine, Scys is serum cystatin C, κ is 0.7 for females and 0.9 for males, α is -0.248 for females and - 0.207 for males, min indicates the minimum of Scr/κ or 1, and max indicates the maximum of Scr/κ or 1.

Table 2. Equations to be used to estimate glomerular filtration rate (GFR)

8 Recording and Reporting of Adverse Events

FAX SAE forms to the eGFR-C Study Office on: 0121 415 9135

Assessment of Safety

There are no novel medical devices or Investigational Medicinal Products (IMPs) used as part of this study. Any study-related Serious Adverse Events (SAEs) which require immediate reporting will be reported on a study-specific SAE form and will follow the procedure/timeframes outlined in this section of the protocol.

Other outcomes, which may also be considered safety outcomes, but which are anticipated outcomes for this group of patients (and therefore not considered studyrelated), will not be reported as adverse events for this study.

Administration of lohexol (i.e. reference method)

The only study-related safety risk to participants involves the administration of iohexol (contrast medium) required for the reference GFR, which does not form part of routine clinical practice and takes approximately five hours to complete. The associated risks *may* include:

- risk from venepuncture and vein cannulation
- idiosyncratic reaction to iohexol
- theoretical risk of deteriorating kidney function with injection of iohexol

These risks are in all cases extremely low. Emergency procedures, if required, will follow standard hospital protocols and as the participants will be in designated clinical areas full resuscitation equipment will be available. The repeated blood tests required will be taken by staff who are experts in phlebotomy. Where possible, a cannula will be left in situ following injection of iohexol, relieving pain or distress from repeated venepuncture.

Adverse Events (AEs)

Generic definitions of different types of AEs are listed in **Table 3**.

Term	Definition				
Adverse Event (AE)	Any untoward medical occurrence in a patient or study subject participating in a study and which does not necessarily have a causal relationship to the study procedure(s)				
Adverse Reaction (AR)	Any untoward and unintended response in a subject to a study procedure where a causal relationship can't be ruled out				
Serious adverse event (SAE), serious adverse reaction (SAR) or unexpected serious adverse reaction	 Any adverse event, adverse reaction or unexpected adverse reaction, respectively, that: 1. results in death; 2. is life-threatening; 3. requires hospitalisation or prolongation of existing hospitalisation; 4. results in persistent or significant disability or incapacity; or 5. consists of a congenital anomaly or birth defect 				

Table 3: AE definitions

It will be left to the Investigator's clinical judgment whether or not an AE is of sufficient severity to require the participant's withdrawal from on-going study participation.

Serious Adverse Events (SAEs)

Disease related morbidity and routine treatment or monitoring of a pre-existing condition that has not worsened will **NOT** be considered as SAEs and should **NOT** be reported to the **eGFR-C** Study Office.

The main theoretically possible recognised reportable SAEs associated with this study relates to the administration of iohexol required for the reference GFR. SAEs occurring within 24 hours of iohexol administration will always be reportable to the **eGFR-C** Study Office on an SAE form. The assessment of relatedness and expectedness to the administration of iohexol is a clinical decision based on all available information at the time.

SAEs outside of this timeframe can also be reported if it is the opinion of the Investigator that there is a possible causal relationship to another aspect of the study.

Assessment of Relatedness

Relatedness of the SAE will be assessed by the Investigator or an authorised member of the site research team (as delegated on the Site Signature and Delegation Log); it must be assessed by a physician. The categories outlined in **Table 4** will be used to define the relatedness (causality) of the SAE.

Category	Definition	Causality
Definitely	There is clear evidence to suggest a causal relationship, and other possible contributing factors can be ruled out	
Probably	There is evidence to suggest a causal relationship, and the influence of other factors is unlikely	
Possibly	There is some evidence to suggest a causal relationship (e.g. the event occurred within a reasonable time after administration of iohexol). However, the influence of other factors may have contributed to the event (e.g. the patient's clinical condition, other concomitant events or medication)	Related
Unlikely	There is little evidence to suggest there is a causal relationship (e.g. the event did not occur within a reasonable time after administration of iohexol). There is another reasonable explanation for the event (e.g. the patient's clinical condition, other concomitant events or medication)	Unrelated
Not related	There is no evidence of any causal relationship	

Table 4: categorisation of relatedness

Assessment of Expectedness

Expectedness will be assessed by the CI or designee using this study protocol as the reference document. **Table 5** gives definitions of expectedness with respect to SAEs.

Table 5: categorisation of expectedness

Category	Definition
Expected	An adverse event that is classed in nature as serious and which is consistent with known information about the study related procedures or that is clearly defined in this protocol
Unexpected	An adverse event that is classed in nature as serious and which is <u>not</u> consistent with known information about the study related procedures

Reporting Timeframes

All SAEs as defined above as requiring reporting on an SAE form must be reported to the **eGFR-C** Study Office within 48 hours of discovery of the event; these will referred to the CI or designee within one working day.

Site Investigators should also report any SAEs to their own institutions as required by their own institutional policies.

SAE Reporting Procedures

Completed SAE forms should be faxed to the **eGFR-C** Study Office on **0121 415 9135** within 48 hours of discovery of the event. The Investigator at each site will be required to respond to any related queries raised by the **eGFR-C** Study Office as soon as possible.

Expedited Reporting to the Research Ethics Committee

Related and Unexpected SAEs

SAEs will be subject to review by the Chief Investigator (or designee).

The Chief Investigator (or designee) will undertake urgent review of all SAEs and may request further information immediately from the clinical team at site. The

Chief Investigator will review the causality and assess the expectedness of the event. Related and Unexpected SAEs will be notified to the MREC by the **eGFR-C** Study Office within 15 days after the Study Office has been notified.

The **eGFR-C** Study Office (on behalf of the Chief Investigator) will report all Related and Unexpected SAEs to the MREC concerned using the standard National Research Ethics Service (NRES) SAE report form for non-CTIMPs.

The **eGFR-C** Study Office (on behalf of the Chief Investigator) will also inform all Investigators concerned of relevant information about SAEs that could adversely affect the safety of participants.

In addition, at regular time points, the Study Steering Committee will be provided with details of all SAEs.

Annual Progress Reports

An annual progress report (with safety information included) will be submitted to the MREC within 30 days of the anniversary date on which the favourable opinion was given, and annually until the study is declared ended. Progress Reports will also be submitted to the Funder in accordance with their requirements.

Reporting urgent safety measures to the Research Ethics Committee

If any urgent safety measures are taken BCTU shall immediately and in any event no later than 3 days from the date the measures are taken, give written notice to the MREC of the measures taken and the circumstances giving rise to those measures.

Notification of Serious Breaches of GCP and/or the protocol

A "serious breach" is a breach which is likely to effect to a significant degree:

- (a) the safety or physical or mental integrity of the participants of the study; or
- (b) the scientific value of the study.

The BCTU on behalf of the Co-Sponsors shall notify the MREC in writing of any serious breach of:

(a) the conditions and principles of GCP in connection with the study; or

(b) the protocol relating to the study, as amended from time to time, within 7 days of becoming aware of that breach.

The Co-Sponsors will be notified immediately of any case where the above definition applies during the study conduct phase.

9 Data Management and Quality Assurance

Birmingham Clinical Trials Unit (BCTU), College of Medical & Dental Sciences, Public Health Building, Edgbaston, Birmingham, B15 2TT Telephone: 0121 415 9134 Fax: 0121 415 9135 Email: eGFR-C@trials.bham.ac.uk

The named clinicians at the participating sites will enter data onto the CRFs. Where this duty is delegated to other staff, this will be recorded in a delegation log. Investigators will keep their own study file logs which link participants with anonymised CRFs.

Data from this study will be handled by the BCTU, a full-time research facility dedicated to, and with substantial experience in, the design and conduct of clinical research. The BCTU recognises the responsibilities of a data management centre with respect to the ethical practice of research and the adequate protection of human subjects.

9.1 Confidentiality of Personal Data

The study will collect personal data about participants, medical records will be reviewed for all patients and routine physical examinations will be performed.

Personal data recorded on all documents will be regarded as strictly confidential and will be handled and stored in accordance with the Data Protection Act 1998. With the patient's consent, their name, NHS number and date of birth will be collected at study entry to assist with long-term follow-up. Patients will be identified using only their unique study number and date of birth in mmm/yyyy format on the CRFs, samples and correspondence between the Study Office and the participating site.

Investigators will keep their own study file logs which link patients with anonymised CRFs. The Investigator must maintain documents not for submission to the Study Office (e.g. Patient Identification Logs) in strict confidence. In the case of specific issues and/or queries from the regulatory authorities, it will be necessary to have access to the complete study records, provided that patient confidentiality is protected.

The Study Office will maintain the confidentiality of all patient data and will not disclose information by which patients may be identified to any third party other than those directly involved in the treatment of the patient and organisations for which the patient has given explicit consent for data transfer. Representatives of the eGFR-C study team may be required to have access to patient's notes for quality assurance purposes but patients should be reassured that their confidentiality will be respected at all times.

The patient consent form, which will be sent to the BCTU will, out of necessity, contain identifiable personal data. These will be stored separately from the study record. The consent form will be sent to BCTU, with the patient's consent, to monitor that the consent process has been completed correctly.

Participants will be informed that their study data and information will be securely stored at the study office at the BCTU, and will be asked to consent to this. The data will be stored on a secure computer database, and all personal information obtained for the study will be held securely and treated as strictly confidential. Any data processed outside of the BCTU will be anonymised.

9.2 Long-Term Storage of Data

In line with the Medicines for Human Use (Clinical Trials) Regulations, once data collection is complete on all participants, all data will be stored for at least 5 years (but ideally not less than 15 years). Any queries or concerns about the data, conduct or conclusions of the study can also be resolved in this time. Limited data on the participants and records of any adverse events may be kept for longer if recommended by an independent advisory board.

Study data will be stored within the BCTU under controlled conditions for at least 3 years after closure. Long-term offsite data archiving facilities will be considered for storage after this time. The BCTU has standard processes for both hard copy and computer database legacy archiving.

9.3 Data collection

When available, case report forms (CRFs) can be entered online at https://www.trials.bham.ac.uk/egfrc. Authorised staff at sites will require an individual secure login username and password to access this online data entry system. Paper CRFs must be completed, signed/dated and returned to the Study Office by the Investigator or an authorised member of the site research team (as delegated on the Site Signature and Delegation Log) within the timeframe listed

above. The exception to this will be the SAE Form which must be co-signed by the Investigator.

Entries on the CRF should be made in ballpoint pen, in blue or black ink, and must be legible. Any errors should be crossed out with a single stroke, the correction inserted and the change initialled and dated. If it is not obvious why a change has been made, an explanation should be written next to the change.

The CRFs will comprise the following forms:

Form	Summary of data recorded	Schedule for submission
Registration Notepad	Patient details (name, NHS number, date of birth, sex and ethnicity), confirmation of eligibility and clinical details. Collected from patient and hospital notes.	Details collected via online registration at baseline.
Baseline and 36 month data collection form	Clinical history and current medication details, anthropometric measurements (height, weight, hip and waist circumference, blood pressure, samples collection, death and withdrawal information. Collected from patient and hospital notes. Form identified by study number and date of birth in mmm/yyyy format.	Faxed/sent/submitted online after each follow-up assessment time point (baseline and 36 months main study, additionally at 12 and 24 months for the disease progression sub-study and at baseline, 2, 3 and 4 weeks for the intra-individual biological variability study)
Serious Adverse Event Form	Details of any SAE occurring within 24 hours of the iohexol testing. Collected from hospital notes. Form identified by study number and date of birth in mmm/yyyy format. Also submitted to study office to record full participant death details.	Faxed within 24 hours of research staff becoming aware of event.
Laboratory Data Collection Form	Results of blood and urine tests, eGFR and GFR results. Collected from patient samples. Form identified by study number and date of birth in mmm/yyyy	Details submitted online and laboratory will retain source data from tests on CRFs and own databases. Samples collected at centres at baseline, 6, 12, 18, 24, 30, 36 months, and baseline, 2, 3 and 4

	format.	weeks for the intra-individual biological variability study. The samples will be sent to the laboratories and analysed in batches (assayed in duplicate for the intra- individual biological variability study).
Follow-up data collection form	Samples collection information, death and withdrawal information. Form identified by study number and date of birth in mmm/yyyy format.	Faxed/sent/submitted online after each follow-up assessment time point (6, 12, 18, 24 and 30 months).

Data reported on each form should be consistent with the source data or the discrepancies should be explained. If information is not known, this must be clearly indicated on the form. All missing and ambiguous data will be queried. All sections are to be completed before returning. In all cases, it remains the responsibility of the Investigator to ensure that the CRF has been completed correctly and that the data are accurate. The completed originals should be sent to the Study Office and a copy filed in the Investigator Site File.

Study forms may be amended by the Study Office, as appropriate, throughout the duration of the study. Whilst this will not constitute a protocol amendment, new versions of the form must be implemented by participating sites immediately on receipt.

All laboratory test results will be entered onto the online system at the central testing laboratories. Authorised staff at the laboratories will require an individual secure login username and password to access this online data entry system. Relevant source data in the laboratories will be stored for a minimum of 10 years.

It will be the responsibility of the investigator to ensure the accuracy of all data entered in the CRFs. The eGFR-C Study Signature & Delegation Log will identify all those personnel with responsibilities for data collection.

9.4 Data handling and analysis

Data analysis will be undertaken by statisticians within BCTU and the University of Birmingham, and by the statistical co-applicant based at the University of Kent. All

data transferred between units will be fully anonymised and transferred in encrypted form.

Personal data and sensitive information required for the eGFR-C study will be collected directly from study participants, their samples and hospital notes on data collection forms, coded with the participant's unique study number and date of birth in the mmm/yyyy format. The consent form will also be faxed to the eGFR-C study office. As this document has identifiable details this will be with consent from the participant. Participants will be informed about the transfer of this information to the eGFR-C study office at the BCTU and asked for their consent. The data will be entered onto a secure computer database, either directly via the internet using secure socket layer encryption technology or indirectly from paper by BCTU staff.

All personal information received in paper format for the study will be held securely and treated as strictly confidential according to BCTU policies. All staff involved in the eGFR-C study (clinical, academic, BCTU) share the same duty of care to prevent unauthorised disclosure of personal information. No data that could be used to identify an individual will be published. Data will be stored on a secure server at Birmingham Clinical Trials Unit (BCTU) under the provisions of the Data Protection Act and/or applicable laws and regulations.

10 Archiving

Archiving will be authorised by the BCTU on behalf of the Co-Sponsors following submission of the end of study report.

Principal Investigators are responsible for the secure archiving of essential study documents (for their site) as per their NHS Trust policy. All essential documents will be archived for a minimum of 5 years after completion of study.

11 Statistical Considerations

11.1 Outcome Measures

11.1.1 Main study

11.1.1.1 Primary outcome measures

 Percentage error in eGFR compared to reference GFR. Values will be computed (for objectives 1 and 2) using baseline data for each of the four eGFR equations, and the proportions within 30% (P₃₀) of the reference standard quoted and compared as a measure of accuracy and precision.

ii) Difference in rate of change in eGFR compared to rate of change in reference GFR. Rates of change will be estimated (for objective 3) using linear regression and the change per annum and percentage change per annum compared to baseline values computed. Large error will be defined as differing by more than 3 mL/min/1.73 m²/year or by 5 percent points per annum.

11.1.1.2 Secondary outcome measures

- Progressive loss of kidney function defined as decline in GFR of more than 10 mL/min/1.73 m²/3 years or increase in albuminuria category as defined by KDIGO (used for objective 4).
- iv) Cost per quality adjusted life year (QALY) (used for objective 6).

11.1.2 Sub-study of patterns of disease progression

 v) Difference in rate of change in eGFR compared to rate of change in reference GFR based on measurements made every 12 months (objective 5).

11.1.3 Additional study of intra-individual biological variation

vi) Within-individual (CV_I) variation and the critical difference (reference change value, RCV) for reference GFR and the estimated GFR equations.

11.2 Sample size

We will recruit 1300 people to the main study, 375 of whom will be part of the sub-study of disease progression. Based on a drop-out rate of 20% this will provide 1000 evaluable participants in the main cohort, and 100 in each of the ethnically defined subgroups. An additional 20 people will be recruited for the study of intra-individual biological variation. Participants in this additional study will not be ineligible for the main study, but it is not expected that they would take part.

1000 evaluable participants will provide over 90% power to detect a difference of 5% in the P_{30} for the main study comparison of the MDRD_{creatinine} equation and the CKD-EPI_{creatinine} and cystatin c equation. The sample size is informed by a simulation study estimating expected differences in P_{30} and rates of change. Power estimates are based on the proportion of statistically significant (P<0.05) results observed in 1000 simulations using values of P_{30} informed by recent studies.^{12, 13, 26, 29, 30} The key difference considered is between a P_{30} of 81% and 86% - a difference considered clinically important and likely to occur with the expected scale of known differences in imprecision between the equations. The calculation is conservative in that it only takes imprecision into account. If there is a systematic bias then the power will be greater than 90%. The sample size of 100 evaluable participants in the ethnic subgroups will allow P_{30} estimates to be reported with 95% confidence interval width of 10%.

Annual rates of change over 3 years will be estimated by linear regression. Two cut-points for errors in rates of change have been proposed: greater than 3 mL/min/1.73 m²/year error and greater than 5 percent point difference per year.⁵⁸ Based on the measurement error corresponding to P_{30} measures of 81% and 86%, the study has over 90% power to detect differences in the proportions with greater than 3 mL/min/1.73 m²/year error and over 80% power to detect differences in proportions with greater than 5% error.

11.3 Statistical analysis

A separate Statistical Analysis Plan for the eGFR-C study provides a detailed description of the planned statistical analyses. A brief outline of these analyses is given below.

11.3.1 Primary outcome analysis

Main study

Data will be analysed to address two main questions: (1) which of the GFRestimating equations is the most accurate assessment of reference GFR; and (2) which most accurately reflects change in GFR? In each case, data will be further analysed to assess whether observed relationships amongst African-Caribbean and South Asian subjects differ from those observed amongst Caucasians, and whether diabetes and proteinuria are predictors.

11.3.1.1 Which of the GFR-estimating equations is the most accurate assessment of reference GFR?

The primary comparison (for which the study is powered) will be between the MDRD_{creatinine} equation (current practice) and the CKD-EPI_{creatinine} and cystatin c equation. The primary analysis will use the baseline data to estimate P_{30} values for all GFR estimating equations and compare P_{30} values between GFR estimating equations using McNemar's test for paired data. An estimate of the difference in P_{30} values together with a 95% confidence interval will be presented. Additional pairwise comparisons will be made between these two equations and the CKD-EPI_{cystatin C} and CKD-EPI_{creatinine} equations.

The mean and median differences between iohexol GFR and each of the estimates of GFR will provide measures of bias.⁶⁴ Differences in precision will be assessed by comparing the interquartile range (IQR) of the estimated GFR. Accuracy will be assessed by establishing the proportion of GFR estimates within 30% (P_{30}) of iohexol GFR, and also as root mean square error (RMSE) of the differences. For bias, IQR, P_{30} and RMSE 95% confidence intervals (CIs) will be estimated using boot-strapping procedures where necessary.

11.3.1.2 Which GFR-estimating equation most accurately reflects change in GFR?

We will compute the average rate of change in measured and estimated GFR across 3 years follow-up, and the difference between them (the error). eGFR data will be modelled as a linear function of time as described by Padala et al⁵⁸ utilising all available (maximum 7) eGFR time points compared to the difference between final and baseline reference GFR values. Rates of change will be expressed both as the absolute regression slopes (standardised per year), and as a percentage change per year comparing the slope to the baseline value.

The percentage of participants demonstrating large error with the respective GFR estimating equations will be compared using McNemar's test.

Large error between measured and estimated GFR will be defined as

- greater than 3 mL/min/1.73 m²/year (primary definition)
- greater than 5% difference in slope (secondary definition)

We will also summarise the ability of the equations to detect changes in GFR over three years defined as

• change in iohexol GFR>10 mL/min/1.73 m² (primary)

- change in iohexol GFR>25% (secondary)
- change in iohexol GFR > the reference change value (RCV) derived in variability study (secondary)

The sensitivity, specificity, positive and negative predictive values and likelihood ratios for the four eGFRs for detecting these changes will be calculated using standard approaches and presented with 95% CIs.

11.3.2 Secondary outcome analysis

11.3.2.1 Which GFR-estimating equation, together with ACR, or ACR alone, most accurately predicts those people that have progressive loss of kidney function (CKD progression)?

Models will be constructed to predict time to progression based on baseline eGFRs and ACR. Progression will be defined in terms of decline in reference GFR (change in iohexol GFR>10 mL/min/1.73 m²) or an increase in albuminuria category, as defined by KDIGO. Progression will only be detected at one of 6 time points, hence piecewise survival models will be fitted to determine whether the prognostic value of ACR and the estimated GFRs is independent of other risk factors. We will develop a prognostic model utilising age, gender, ethnicity, BMI, waist circumference, MABP, diabetes mellitus, smoking status, and presence of vascular disease in addition to baseline ACR and the various eGFRs. Both proportional and non-proportional hazards will be considered. Bootstrap validation will be used with these prediction models.

11.3.2.2 Cost-effectiveness modelling of the optimal strategy

Whilst our longitudinal cohort will not have adequate power to detect differences in progression, our estimates of the accuracy of eGFR (main study), patterns and determinants of progression (sub-study), and intraindividual biological variation (additional study) can be combined in a model to evaluate the impact of alternative monitoring strategies on detection of progression to stage 4 CKD. True GFR values will be modelled over time for representative cohorts of people, and performance of alternative monitoring strategies in detecting progression to stage 4 CKD (varying in timing and choice of eGFR equation) will be simulated utilising estimates of measurement error and accuracy. Outcome variables which will be assessed will include sensitivity, false positive progression rates, and delays in detecting progression. These modelling approaches can then contribute to the economic evaluation to establish the most cost-effective monitoring strategy. A decision analytic model will be implemented which will incorporate the information describing the performance of the alternative monitoring strategies, and will consider the impact of the error in eGFR measures on patient outcomes. Data on the type and frequency of tests for the alternative monitoring strategies is being collected over time, along with the description and frequency of any adverse events, and these will help to inform the costs and outcomes of each of the monitoring strategies. Data describing long term patient outcomes that may occur beyond the end of the study will be obtained from secondary sources. This will enable the impact of the error in the eGFR measures on patient outcomes such as myocardial infarction, the need for kidney transplant, and renal failure to be incorporated into the analysis.

This cost-effectiveness analysis will take the form of a cost-utility analysis in which the outcome measure will be the cost per quality adjusted life year (QALY). In the short term the alternative testing strategies are unlikely to impact on the quality of life (QoL) of the patients over the period of the study (hence why no QoL data is being collected in this study); however failure to detect the progression of CKD may impact on later patient outcomes if the appropriate treatment is not administered in a timely manner. The cost and QoL of these later patient outcomes will be informed by secondary sources, and will be incorporated into the economic evaluation.

In the first instance prior to the economic evaluation being undertaken a systematic review of the cost-effectiveness literature on CKD will be implemented. This will inform as to the types of approaches to the modelling of CKD that have been undertaken in the past, and will also help to identify the parameters that will be necessary to extrapolate the model beyond the time horizon of the study.

11.3.2.3 Sub-study of patterns of disease progression - How does GFR progress over time and what are the optimal monitoring times?

The rate of decline (mL/min/1.73 m²/year) in reference GFR, and the difference between reference GFR and estimated GFRs (referred to as error), measured every 12 months will be modelled over time using a longitudinal linear or nonlinear (exponential decline) random coefficients regression model to estimate average and variability in disease progression and error. Parameters of the model for each outcome will be estimated using maximum likelihood. The fixed effect component of the models will adjust the expected value of the population average to take account of significant covariates. Outcomes will be modelled on the natural and log-transformed scales. Covariates to be explored in the model will include gender, age, diabetes, duration of diabetes, ethnicity, albuminuria, baseline GFR, blood pressure,

BMI, waist circumference, smoking status and presence of vascular disease. The effect of covariates on the population average intercept and longitudinal time effect parameters will be assessed. The method of backward elimination will be used to remove covariates which are not significant from the model. Between and within patient variability in the rate of decline of reference GFR will also be estimated. Linear relationships between disease progression and drug name or drug class will be explored.

Analysis will be undertaken using NONMEM version 7.1.2, R open source software and PFIM version 3.2.2 optimal design algorithms (R open source software). The PFIM algorithms will be used to calculate the D-optimal⁶⁵ sampling times from the disease progression model based on reference GFR for people with diabetes and/or proteinuria, and for those people with neither of these conditions. The point estimates and the covariance matrix of model estimates will be used in the optimal design algorithms to ensure the solution is robust to model uncertainty. Optimal monitoring strategies will be selected from a set of designs with sampling every six months, and monitoring strategies with a number of sampling points (between two and six) will be compared. Optimal monitoring strategies will be calculated for other important subgroups of the population found to be significant in the disease progression model. The monitoring strategies identified will be used as the basis for further simulations.

Additional disease progression modelling will be performed on eGFRs and ACR. Parameter estimates and estimates of within- and between-patient variability will be compared to those for reference GFR.

11.3.2.4 Additional study of intra-individual biological variation

Pre-analytical variables will be standardised. Participants completing less than 2 of the 4 planned reference test procedures will be excluded from analysis. All samples for all analytes will be assayed in duplicate and the analytical variance (SD_A^2) will be calculated from the differences between the duplicate measurements. The data will be initially examined and the requirement to normalise the data using a logarithmic transformation will be considered. Outliers will be excluded using Cochran's test and Reed's test, as described by Fraser and Harris.²² The total (CV_T), analytical (CV_A), group (CV_G) and within-individual (CV_I) components of variation will be calculated using nested ANOVA.²² The critical difference (reference change value, RCV) for significant changes in serial results (P<0.05) and the number of specimens required to estimate the homeostatic set-point of an individual (within ±10% with a confidence of 95%) will also be estimated. The derived RCV for the reference

GFR will be used to test the ability of estimated GFR equations to detect a true change in GFR (see 12.3.1.2 above).

11.4 Interim analyses

Interim analyses will focus on ensuring that the recruited sample (both overall and in the sub-study of disease progression) reflects the targeted groups.

Targets for recruitment of participants with proteinuria (ACR >30 mg/mmol) and diabetes will ensure that at least 20% and no more than 80% are within these groups.

Targets for the African-Caribbean and South Asian groups are for proportions of 10-15% in the overall study, and 125 of each recruited to the disease progression substudy.

At regular intervals during the recruitment phase the recruitment pattern will be examined against these metrics and adjustments made to the recruitment process if required.

11.5 Final analyses

There are five stages of analysis in the study:

- The additional study of biological variation (12.3.2.4) will be analysed as soon as all 20 participants have completed and test results are available (expected to be in Year 1).
- 2) The analysis of accuracy of the equations for measuring eGFR (12.3.1.1) will be based on analysis of baseline values, and will be completed once baseline test results are available on all participants (expected to be in Year 4)
- The modelling of GFR progression over time (12.3.2.3) will be undertaken after the last participant in the disease progression sub-study has undergone the final reference GFR assessment.
- 4) The final analysis of the accuracy of change in eGFR (12.3.1.2 and 12.3.2.1) will be undertaken after the last participant has undergone their final reference GFR assessment (expected to be in Year 7).
- 5) The health economic analysis will be undertaken when the final analysis of accuracy of change is complete (12.3.2.2).

11.6 End of study

The end of study will be 6 months after the last data capture. The last data capture will be 36 months following recruitment of the last participant.

12 Direct Access to Source Data

The investigator(s)/institution(s) will permit study-related monitoring, audits and REC review, providing direct access to source data/documents. Study participants are informed of this during the informed consent discussion and will consent to provide access to their medical notes.

13 Ethics Requirements

The Co-Sponsors will ensure that the study protocol, PIS, consent form, GP letter and submitted supporting documents have been approved by the MREC prior to any participant recruitment. The protocol and all agreed substantial protocol amendments will be documented and submitted for ethical approval prior to implementation.

Before a site can enrol participants into the study, the Principal Investigator or designee must apply for NHS permission from their Trust Research & Development (R&D) and be granted written permission. It is the responsibility of the Principal Investigator or designee at each site to ensure that all subsequent amendments gain the necessary approval. This does not affect the individual clinician's responsibility to take immediate action if thought necessary to protect the health and interest of individual participants.

Within 90 days after the end of the study, the Chief Investigator/ Co-Sponsors will ensure that the MREC is notified that the study has finished. If the study is terminated prematurely, those reports will be made within 15 days after the end of the study.

The Chief Investigator will supply the Co-Sponsors with a summary report of the clinical study, which will then be submitted to the MREC within one year after the end of the study.

14 Monitoring Requirement for the Study

Monitoring of this study will be to ensure compliance with GCP. A risk proportionate approach to the initiation, management and monitoring of the study will be adopted (as per the MRC/DH/MHRA Joint Project: Risk-adapted Approaches to the Management of Clinical Studies of Investigational Medicinal Products) and outlined in the study-specific risk assessment.

There will be an independent study steering committee (SSC). This will comprise an independent chairperson (Dr Charlie Tomson, Consultant Nephrologist, Bristol), one other independent nephrologist, a patient representative, the study statistician, an independent statistician, the chief investigator, the study lead research nurse and one co-applicant. This group will meet at the beginning of the study and thereafter up to six monthly depending on progress.

The study will not have an independent data monitoring committee.

15 Finance

The National Institute for Health Research is funding this study through its Health Technology Assessment programme.

16 Indemnity

This is a clinician-initiated study, ABPI guidelines for patient compensation by the pharmaceutical industry will not apply.

The Co-Sponsors hold Public Liability (negligent harm) and Clinical Trial (negligent harm) insurance policies, which apply to this study. Participants may be able to claim compensation, if they can prove that either Co-Sponsor has been negligent. However, as this study is being carried out in a hospital setting, NHS Trust and Non-Trust Hospitals have a duty of care to patients treated, whether or not the patient is taking part in a clinical study. Compensation is only available via NHS indemnity in the event of clinical negligence being proven. The Co-Sponsors do not accept liability for any breach in the hospital's duty of care, or any negligence on the part of hospital employees.

Participants *may* also be able to claim compensation for injury caused by participation in this study without the need to prove negligence on the part of the Co-Sponsors or another party. Participants who sustain injury and wish to make a claim for compensation should do so in writing in the first instance to the Chief Investigator, who will pass the claim to the relevant Insurers.

There are no specific arrangements for compensation made in respect of any serious adverse events occurring though participation in the study, whether from the side effects listed, or others yet unforeseen.

Hospitals selected to participate in this study shall provide clinical negligence insurance cover for harm caused by their employees and a copy of the relevant insurance policy or summary should be provided to Co-Sponsors, upon request.

17 Dissemination and Publication

The Chief Investigator will coordinate dissemination of data from this study. All publications and presentations, including abstracts, relating to the study will be authorised by the eGFR-C Study Management Group. The results of the analysis will be published in a peer reviewed journal either in the name of all of the members of the Study Management Group listed individually or in the name of the eGFR-C Collaborative Study Group collectively in a peer reviewed journal (provided that this does not conflict with the individual journal's policy). In the latter case all contributors to the study will be listed, with their contribution identified. Study participants will be sent a summary of the final results of the study, which will contain a reference to the full paper.

All publications using data from this study to undertake original analyses will be submitted to the Study Management Group for review before release. To safeguard the scientific integrity of the study, data will not be presented in public before the main results are published without the prior consent of the Study Management Group.

18 Statement of Compliance

The study will be conducted in compliance with the approved protocol, EU GCP and the Research Governance Framework.

19 References

- 1. Carter JL, Stevens PE, Irving JE, Lamb EJ. Estimating glomerular filtration rate: comparison of the CKD-EPI and MDRD equations in a large UK cohort with particular emphasis on the effect of age. *QJM* 2011;104(10):839-47.
- Coresh J, Astor BC, Greene T, Eknoyan G, Levey AS. Prevalence of chronic kidney disease and decreased kidney function in the adult US population: Third National Health and Nutrition Examination Survey. *Am J Kidney Dis* 2003;41(1):1-12.
- Stevens PE, O'Donoghue DJ, de Lusignan S, Van Vlymen J, Klebe B, Middleton R, et al. Chronic kidney disease management in the United Kingdom: NEOERICA project results. *Kidney Int* 2007;72(1):92-9.
- Roth M, Roderick P, Mindell J. Chapter 3: Kidney Disease and renal function in Health Survey for England 2009: NHS Information Centre, 2010 Available from: <u>http://www.ic.nhs.uk/statistics-and-data-collections/health-and-lifestylesrelated-surveys/health-survey-for-england</u>.
- Gaspari F, Perico N, Ruggenenti P, Mosconi L, Amuchastegui CS, Guerini E, et al. Plasma clearance of nonradioactive iohexol as a measure of glomerular filtration rate. *J Am Soc Nephrol* 1995;6(2):257-63.
- 6. Department of Health. National Service Framework for Renal Services. Part Two: Chronic Kidney Disease, Acute Renal Failure and End of Life Care. 2005.
- National Institute for Health and Care Excellence. Chronic Kidney Disease: National Clinical Guideline for Early Identification and Management in Adults in Primary and Secondary Care. Clinical Guideline 73. 2008.
- de Lusignan S, Tomson C, Harris K, van Vlymen J, Gallagher H. Creatinine fluctuation has a greater effect than the formula to estimate glomerular filtration rate on the prevalence of chronic kidney disease. *Nephron Clin Pract* 2011;117(3):c213-24.
- 9. Levin A, Stevens PE. Early detection of CKD: the benefits, limitations and effects on prognosis. *Nat Rev Nephrol* 2011;7(8):446-57.
- 10. Moynihan R, Doust J, Henry D. Preventing overdiagnosis: how to stop harming the healthy. *Bmj* 2012;344:e3502.
- Kidney Disease Improving Global Outcomes. Clinical Practice Guideline for the Evaluation and Management of Chronic Kidney Disease. *Kidney International* 2013;3: (Suppl.):1-150.
- Levey AS, Bosch JP, Lewis JB, Greene T, Rogers N, Roth D. A more accurate method to estimate glomerular filtration rate from serum creatinine: a new prediction equation. Modification of Diet in Renal Disease Study Group. *Ann Intern Med* 1999;130(6):461-70.
- 13. Levey AS, Stevens LA, Schmid CH, Zhang YL, Castro AF, 3rd, Feldman HI, et al. A new equation to estimate glomerular filtration rate. *Ann Intern Med* 2009;150(9):604-12.

- 14. Hsu CY, Bansal N. Measured GFR as "gold standard"--all that glitters is not gold? *Clin J Am Soc Nephrol* 2011;6(8):1813-4.
- Brochner-Mortensen J, Rodbro P. Selection of routine method for determination of glomerular filtration rate in adult patients. *Scand J Clin Lab Invest* 1976;36(1):35-43.
- Gaspari F, Perico N, Matalone M, Signorini O, Azzollini N, Mister M, et al. Precision of plasma clearance of iohexol for estimation of GFR in patients with renal disease. *J Am Soc Nephrol* 1998;9(2):310-3.
- 17. Levey AS, Greene T, Schluchter MD, Cleary PA, Teschan PE, Lorenz RA, et al. Glomerular filtration rate measurements in clinical trials. Modification of Diet in Renal Disease Study Group and the Diabetes Control and Complications Trial Research Group. J Am Soc Nephrol 1993;4(5):1159-71.
- Wilkinson J, Fleming JS, Waller DG. Effect of food and activity on the reproducibility of isotopic GFR estimation. *Nucl Med Commun* 1990;11(10):697-700.
- Blake GM, Roe D, Lazarus CR. Long-term precision of glomerular filtration rate measurements using 51Cr-EDTA plasma clearance. *Nucl Med Commun* 1997;18(8):776-84.
- Perrone RD, Steinman TI, Beck GJ, Skibinski CI, Royal HD, Lawlor M, et al. Utility of radioisotopic filtration markers in chronic renal insufficiency: simultaneous comparison of 125I-iothalamate, 169Yb-DTPA, 99mTc-DTPA, and inulin. The Modification of Diet in Renal Disease Study. *Am J Kidney Dis* 1990;16(3):224-35.
- 21. Kwong YT, Stevens LA, Selvin E, Zhang YL, Greene T, Van Lente F, et al. Imprecision of urinary iothalamate clearance as a gold-standard measure of GFR decreases the diagnostic accuracy of kidney function estimating equations. *Am J Kidney Dis* 2010;56(1):39-49.
- 22. Fraser CG, Harris EK. Generation and application of data on biological variation in clinical chemistry. *Crit Rev Clin Lab Sci* 1989;27(5):409-37.
- Levey AS, Greene T, Kusek JW, Beck GJ. A simplified equation to predict glomerular filtration rate from serum creatinine. J Am Soc Nephrol 2000;11(suppl):155A.
- Levey AS, Coresh J, Greene T, Stevens LA, Zhang YL, Hendriksen S, et al. Using standardized serum creatinine values in the modification of diet in renal disease study equation for estimating glomerular filtration rate. *Ann Intern Med* 2006;145(4):247-54.
- 25. Myers GL, Miller WG, Coresh J, Fleming J, Greenberg N, Greene T, et al. Recommendations for improving serum creatinine measurement: a report from the Laboratory Working Group of the National Kidney Disease Education Program. *Clin Chem* 2006;52(1):5-18.
- Earley A, Miskulin D, Lamb EJ, Levey AS, Uhlig K. Estimating Equations for Glomerular Filtration Rate in the Era of Creatinine Standardization: A Systematic Review. Ann Intern Med 2012;156:785-795.

- 27. Dharnidharka VR, Kwon C, Stevens G. Serum cystatin C is superior to serum creatinine as a marker of kidney function: a meta-analysis. *Am J Kidney Dis* 2002;40(2):221-6.
- 28. Laterza OF, Price CP, Scott MG. Cystatin C: an improved estimator of glomerular filtration rate? *Clin Chem* 2002;48(5):699-707.
- 29. Inker LA, Schmid CH, Tighiouart H, Eckfeldt JH, Feldman HI, Greene T, et al. A new equation to estimate glomerular filtration rate from standardized creatinine and cystatin C. *N Engl J Med* 2012;367(1):20-29.
- 30. Kilbride HS, Stevens PE, Eaglestone G, Knight S, Carter JL, Delaney MP, et al. Accuracy of the MDRD (Modification of Diet in Renal Disease) study and CKD-EPI (CKD Epidemiology Collaboration) equations for estimation of GFR in the elderly. *Am J Kidney Dis* 2012;61(1):57-66.
- 31. Raleigh VS. Diabetes and hypertension in Britain's ethnic minorities: implications for the future of renal services. *Bmj* 1997;314(7075):209-13.
- 32. Chandie Shaw PK, Baboe F, van Es LA, van der Vijver JC, van de Ree MA, de Jonge N, et al. South-Asian type 2 diabetic patients have higher incidence and faster progression of renal disease compared with Dutch-European diabetic patients. *Diabetes Care* 2006;29(6):1383-5.
- 33. Lewis J, Agodoa L, Cheek D, Greene T, Middleton J, O'Connor D, et al. Comparison of cross-sectional renal function measurements in African Americans with hypertensive nephrosclerosis and of primary formulas to estimate glomerular filtration rate. *Am J Kidney Dis* 2001;38(4):744-53.
- 34. Ma YC, Zuo L, Chen JH, Luo Q, Yu XQ, Li Y, et al. Modified glomerular filtration rate estimating equation for Chinese patients with chronic kidney disease. J Am Soc Nephrol 2006;17(10):2937-44.
- 35. Matsuo S, Imai E, Horio M, Yasuda Y, Tomita K, Nitta K, et al. Revised equations for estimated GFR from serum creatinine in Japan. *Am J Kidney Dis* 2009;53(6):982-92.
- 36. Horio M, Imai E, Yasuda Y, Watanabe T, Matsuo S. Modification of the CKD epidemiology collaboration (CKD-EPI) equation for Japanese: accuracy and use for population estimates. *Am J Kidney Dis* 2010;56(1):32-8.
- 37. Stevens LA, Schmid CH, Greene T, Zhang YL, Beck GJ, Froissart M, et al. Comparative performance of the CKD Epidemiology Collaboration (CKD-EPI) and the Modification of Diet in Renal Disease (MDRD) Study equations for estimating GFR levels above 60 mL/min/1.73 m2. Am J Kidney Dis 2010;56(3):486-95.
- Teo BW, Xu H, Wang D, Li J, Sinha AK, Shuter B, et al. GFR estimating equations in a multiethnic Asian population. *Am J Kidney Dis* 2011;58(1):56-63.
- Praditpornsilpa K, Townamchai N, Chaiwatanarat T, Tiranathanagul K, Katawatin P, Susantitaphong P, et al. The need for robust validation for MDRD-based glomerular filtration rate estimation in various CKD populations. *Nephrol Dial Transplant* 2011;26(9):2780-5.

- Barbour SJ, Er L, Djurdjev O, Karim M, Levin A. Differences in progression of CKD and mortality amongst Caucasian, Oriental Asian and South Asian CKD patients. *Nephrol Dial Transplant* 2010;25(11):3663-72.
- Barbour SJ, Schachter M, Er L, Djurdjev O, Levin A. A systematic review of ethnic differences in the rate of renal progression in CKD patients. *Nephrol Dial Transplant* 2010;25(8):2422-30.
- 42. Brazy PC, Fitzwilliam JF. Progressive renal disease: role of race and antihypertensive medications. *Kidney Int* 1990;37(4):1113-9.
- 43. Keith DS, Nichols GA, Gullion CM, Brown JB, Smith DH. Longitudinal follow-up and outcomes among a population with chronic kidney disease in a large managed care organization. *Arch Intern Med* 2004;164(6):659-63.
- 44. Hemmelgarn BR, Zhang J, Manns BJ, Tonelli M, Larsen E, Ghali WA, et al. Progression of kidney dysfunction in the community-dwelling elderly. *Kidney Int* 2006;69(12):2155-61.
- 45. Ruggenenti P, Perna A, Remuzzi G. ACE inhibitors to prevent end-stage renal disease: when to start and why possibly never to stop: a post hoc analysis of the REIN trial results. Ramipril Efficacy in Nephropathy. J Am Soc Nephrol 2001;12(12):2832-7.
- Gall MA, Nielsen FS, Smidt UM, Parving HH. The course of kidney function in type 2 (non-insulin-dependent) diabetic patients with diabetic nephropathy. *Diabetologia* 1993;36(10):1071-8.
- 47. MacIsaac RJ, Tsalamandris C, Panagiotopoulos S, Smith TJ, McNeil KJ, Jerums G. Nonalbuminuric renal insufficiency in type 2 diabetes. *Diabetes Care* 2004;27(1):195-200.
- 48. Parving HH, Andersen AR, Smidt UM, Hommel E, Mathiesen ER, Svendsen PA. Effect of antihypertensive treatment on kidney function in diabetic nephropathy. *Br Med J (Clin Res Ed)* 1987;294(6585):1443-7.
- 49. Bjorck S, Nyberg G, Mulec H, Granerus G, Herlitz H, Aurell M. Beneficial effects of angiotensin converting enzyme inhibition on renal function in patients with diabetic nephropathy. *Br Med J (Clin Res Ed)* 1986;293(6545):471-4.
- 50. Hoefield RA, Kalra PA, Baker PG, Sousa I, Diggle PJ, Gibson MJ, et al. The use of eGFR and ACR to predict decline in renal function in people with diabetes. *Nephrol Dial Transplant* 2011;26(3):887-92.
- 51. Gowans EM, Fraser CG. Biological variation of serum and urine creatinine and creatinine clearance: ramifications for interpretation of results and patient care. *Ann Clin Biochem* 1988;25:259-63.
- 52. Keevil BG, Kilpatrick ES, Nichols SP, Maylor PW. Biological variation of cystatin C: implications for the assessment of glomerular filtration rate. *Clin Chem* 1998;44(7):1535-1539.
- 53. Reinhard M, Erlandsen EJ, Randers E. Biological variation of cystatin C and creatinine. *Scand J Clin Lab Invest* 2009;69(8):831-6.
- 54. Delanaye P, Cavalier E, Depas G, Chapelle JP, Krzesinski JM. New data on the intraindividual variation of cystatin C. *Nephron Clin Pract* 2008;108(4):c246-8.

- 55. Andersen TB, Erlandsen EJ, Frokiaer J, Eskild-Jensen A, Brochner-Mortensen J. Comparison of within- and between-subject variation of serum cystatin C and serum creatinine in children aged 2-13 years. *Scand J Clin Lab Invest* 2010;70(1):54-59.
- 56. Bandaranayake N, Ankrah-Tetteh T, Wijeratne S, Swaminathan R. Intraindividual variation in creatinine and cystatin C. *Clin Chem Lab Med* 2007;45(9):1237-9.
- 57. Toffaletti JG, McDonnell EH. Variation of serum creatinine, cystatin C, and creatinine clearance tests in persons with normal renal function. *Clin Chim Acta* 2008;395(1-2):115-9.
- 58. Padala S, Tighiouart H, Inker LA, Contreras G, Beck GJ, Lewis J, et al. Accuracy of a GFR estimating equation over time in people with a wide range of kidney function. *Am J Kidney Dis* 2012;60(2):217-24.
- 59. Peralta CA, Shlipak MG, Judd S, Cushman M, McClellan W, Zakai NA, et al. Detection of chronic kidney disease with creatinine, cystatin C, and urine albumin-to-creatinine ratio and association with progression to end-stage renal disease and mortality. *JAMA* 2011;305(15):1545-52.
- Waheed S, Matsushita K, Astor BC, Hoogeveen RC, Ballantyne C, Coresh J. Combined association of creatinine, albuminuria, and cystatin C with all-cause mortality and cardiovascular and kidney outcomes. *Clin J Am Soc Nephrol* 2013;8(3):434-442.
- 61. Waheed S, Matsushita K, Sang Y, Hoogeveen R, Ballantyne C, Coresh J, et al. Combined association of albuminuria and cystatin C-based estimated GFR with mortality, coronary heart disease, and heart failure outcomes: the Atherosclerosis Risk in Communities (ARIC) Study. *Am J Kidney Dis* 2012;60(2):207-16.
- 62. Erley CM, Bader BD, Berger ED, Vochazer A, Jorzik JJ, Dietz K, et al. Plasma clearance of iodine contrast media as a measure of glomerular filtration rate in critically ill patients. *Crit Care Med* 2001;29(8):1544-50.
- 63. Schwartz GJ, Work DF. Measurement and estimation of GFR in children and adolescents. *Clin J Am Soc Nephrol* 2009;4(11):1832-43.
- National Kidney Foundation. K/DOQI clinical practice guidelines for chronic kidney disease: evaluation, classification, and stratification. *Am J Kidney Dis* 2002;39(2 Suppl 1):S1-266.
- 65. O'Brien TE, Funk GM. A gentle introduction to optimal design for regression models. *The American Statistician* 2003;57(4):265-267.